Author's response to reviews

Title: Antitumor effectiveness of different amounts of electrical charge in Ehrlich and fibrosarcoma Sa-37 tumors

Authors:

Hector C Ciria (camue@cnea.uo.edu.cu)
Magaly S Quevedo (camue@cnea.uo.edu.cu)
Luis B Cabrales (berquesc@yahoo.com)
Rodolfo P Bruzon (naogit@yahoo.com)
Miriam F Salas (camue@cnea.uo.edu.cu)
Odalis G Pena (amandaernesto@yahoo.es)
Tamara R Gonzalez (camue@cnea.uo.edu.cu)
Demetrio S Lopez (camue@cnea.uo.edu.cu)
Jose M Flores (jmarin@cajeme.cifus.uson.mx)

Version: 4 Date: 13 November 2004

Author's response to reviews:

Author list - The list of authors in the manuscript was written exactly as they are in the submission system, including the usage of middle names.

Figures - All Figures has been submitted as JPEG files

Figure legends - are now only in black and white.